Author’s response to reviews

Title: More is less, less is more, or does it really matter? The curious case of impact of azacitidine administration schedules on outcomes in patients with myelodysplastic syndromes

Authors:

Rory Shallis (rory.shallis@yale.edu)

Amer Zeidan (amer.zeidan@yale.edu)

Version: 2 Date: 13 Nov 2017

Author’s response to reviews:

1. Please include the titles Background, Main text and Conclusions in your manuscript as per our submission guidelines

ANSWER: Background, Main text, and Conclusions titles have been added.

2. The individual contributions of ALL authors to the manuscript should be specified in the "Authors’ Contributions" section, please also specifically include AZ in this section.

ANSWER: All authors have been specified.

3. There is no need to supply a cover letter at this stage.

ANSWER: Cover Letter has been removed.